# Assistive technology and telecare to maintain independent living at home for people with dementia: the ATTILA RCT

Rebecca Gathercole,<sup>1\*</sup> Rosie Bradley,<sup>2</sup> Emma Harper,<sup>2</sup> Lucy Davies,<sup>2</sup> Lynn Pank,<sup>2</sup> Natalie Lam,<sup>2</sup> Anna Davies,<sup>3,4</sup> Emma Talbot,<sup>5</sup> Emma Hooper,<sup>6,7</sup> Rachel Winson,<sup>8</sup> Bethany Scutt,<sup>1</sup> Victoria Ordonez Montano,<sup>9</sup> Samantha Nunn,<sup>10</sup> Grace Lavelle,<sup>1</sup> Matthew Lariviere,<sup>11</sup> Shashivadan Hirani,<sup>3</sup> Stefano Brini,<sup>3</sup> Andrew Bateman,<sup>12</sup> Peter Bentham,<sup>13</sup> Alistair Burns,<sup>7</sup> Barbara Dunk,<sup>14</sup> Kirsty Forsyth,<sup>15</sup> Chris Fox,<sup>16</sup> Catherine Henderson,<sup>17</sup> Martin Knapp,<sup>17</sup> Iracema Leroi,<sup>18</sup> Stanton Newman,<sup>3</sup> John O'Brien,<sup>19</sup> Fiona Poland,<sup>20</sup> John Woolham,<sup>21</sup> Richard Gray<sup>2</sup> and Robert Howard<sup>22</sup>

- <sup>1</sup>Department of Old Age Psychiatry, King's College London, London, UK
- <sup>2</sup>Medical Research Council Population Health Research Unit, University of Oxford, Oxford, UK
- <sup>3</sup>School of Health Sciences, City, University of London, London, UK
- <sup>4</sup>Population Health Sciences, University of Bristol, Bristol, UK
- <sup>5</sup>Norfolk and Suffolk NHS Foundation Trust, Stowmarket, UK
- <sup>6</sup>Lancashire Care NHS Foundation Trust, Preston, UK
- <sup>7</sup>Faculty of Biology, Medicine and Health, University of Manchester, Manchester, UK
- <sup>8</sup>Cambridgeshire and Peterborough NHS Foundation Trust, Cambridge, UK
- <sup>9</sup>Hertfordshire Community NHS Trust, Watford, UK
- <sup>10</sup>Cambridgeshire Community Services NHS Trust, Cambridge, UK
- <sup>11</sup>Centre for International Research on Care, Labour and Equalities, University of Sheffield, Sheffield, UK
- <sup>12</sup>School of Health and Social Care, University of Essex, Colchester, UK
- <sup>13</sup>Birmingham and Solihull Mental Health NHS Foundation Trust, Birmingham, UK
- <sup>14</sup>South London and Maudsley NHS Foundation Trust, London, UK
- <sup>15</sup>School of Health Sciences, Queen Margaret University, Edinburgh, UK
- <sup>16</sup>Norwich Medical School, University of East Anglia, Norwich, UK
- <sup>17</sup>Care Policy and Evaluation Centre, London School of Economics and Political Science, London, UK
- <sup>18</sup>Global Brain Health Institute, Trinity College Dublin, Dublin, Ireland
- <sup>19</sup>Department of Psychiatry, University of Cambridge, Cambridge, UK
- <sup>20</sup>School of Health Sciences, University of East Anglia, Norwich, UK
- <sup>21</sup>National Institute for Health Research (NIHR) Health & Social Care Workforce Research Unit, King's College London, London, UK
- <sup>22</sup>Division of Psychiatry, University College London, London, UK

\*Corresponding author rebecca.gathercole@kcl.ac.uk

**Declared competing interests of authors:** Samantha Nunn reports that a National Institute for Health Research (NIHR) Health Technology Assessment (HTA) programme grant was awarded to King's College London, and that study activity at Cambridgeshire Community Services NHS Trust was funded by this grant. Peter Bentham reports personal fees from TauRx Therapeutics Ltd (Aberdeen, UK) outside the submitted work. Catherine Henderson reports grants from the NIHR HTA programme during the conduct of the study. John O'Brien reports personal fees from TauRx Therapeutics Ltd, AXON Neuroscience SE (Larnaca, Cyprus), GE Healthcare (Chicago, IL, USA) and Eisai Co., Ltd (Tokyo, Japan), grants and personal fees from Avid Radiopharmaceuticals (Eli Lilly and Company, Indianapolis, IN, USA) and grants from Alliance Medical (Warwick, UK) outside the submitted work. Richard Gray reports grants from the NIHR HTA programme during the conduct of the study. He also reports that he was a member of the HTA programme Commissioning Sub-board (Expressions of Interest) (2009–11) and of the HTA Programme Commissioning Committee (2013–18).

Published March 2021 DOI: 10.3310/hta25190

## **Plain English summary**

The ATTILA RCT Health Technology Assessment 2021; Vol. 25: No. 19 DOI: 10.3310/hta25190

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

M any people with dementia living at home are recommended assistive technology and telecare to help them remain living safely and independently in the community. These devices are meant to assist and support activities such as taking medication or cooking, or to raise an alert when there is an issue, such as a fire; however, there is currently little evidence to support such claims. This trial investigated whether or not assistive technology and telecare could delay people moving into residential care and keep them any safer than alternatives, and whether or not they were cost-effective.

We recruited 495 people with dementia and their unpaid caregivers, who were randomly assigned to receive either a package of assistive technology and telecare recommended by a health or social care professional or alternative support involving only basic assistive technology and telecare. We monitored the residential status, the use of health-care services and the health and well-being of participants with dementia and their caregivers over a 2-year period. Researchers also spent time with participants to see how they were living with the technology.

The trial found no difference in the time that people with dementia with full assistive technology and telecare remained at home, nor any reduction in the number of safety incidents, compared with the participants who received basic assistive technology and telecare only. Full assistive technology and telecare did not increase health and social care costs. It did not improve the well-being of people with dementia or that of their caregivers. People with dementia who had full assistive technology and telecare rated their quality of life poorer than those with basic assistive technology and telecare did, but their caregivers rated their quality of life as about the same as caregivers of people with basic assistive technology and telecare. The technology sometimes averted crises but also disrupted people's everyday lives.

These results suggest that assistive technology and telecare for people with dementia provided in real-world conditions may not be as beneficial as previously claimed. The way that assistive technology and telecare services are organised bears further investigation to see how these services could be improved.

### **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA programme**

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

#### **This report**

The research reported in this issue of the journal was funded by the HTA programme as project number 10/50/02. The contractual start date was in January 2013. The draft report began editorial review in September 2019 and was accepted for publication in December 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

Copyright © 2021 Gathercole *et al.* This work was produced by Gathercole *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### NIHR Journals Library Editors

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk